Author’s response to reviews

Title: Polysaccharopeptide enhanced the anti-cancer effect of gamma-tocotrienol through activation of AMPK.

Authors:

Ji Liu (ji.liu@qut.edu.au)
Eunice Yuen-Ting Lau (elkuuga@yahoo.com.hk)
Jiezhong Chen (j.chen4@uq.edu.au)
Joan Yong (wolfjay89@gmail.com)
Kai Dun Tang (kaidun.tang@student.qut.edu.au)
Jessica Lo (finityeiht@gmail.com)
Irene Oi-Lin Ng (iolng@hku.hk)
Terence Kin Wah Lee (tkwlee@hkucc.hku.hk)
Ming Tat Ling (mingtat.ling@qut.edu.au)

Version: 2
Date: 26 June 2014

Author’s response to reviews:

Dear Editor,

Thankyou for your email and the following changes have been made:
1. Line and page numbers have been added.
2. Acknowledgment has been included.
3. Ethics statement has been revised.
4. Competing interests statement has been revised.

Yours,

Patrick